Author’s response to reviews

Title: The attenuation of neurological injury from the use of simvastatin after spinal cord ischemia-reperfusion injury in rats

Authors:

Jung-Hee Ryu (jinaryu74@gmail.com)
Jin-woo Park (jinul8282@gmail.com)
Jin-Young Hwang (mistyblue15@naver.com)
Seong- Joo Park (struka@snubh.org)
Jin-Hee Kim (anesing1@snubh.org)
Hye-Min Sohn (shm@snubh.org)
SUNGHEE HAN (noninvasive@hanmail.net)

Version: 4 Date: 18 Feb 2018

Author’s response to reviews:

Dear Pf Zhu We would like to thank you for the potential acceptable for the publication in BMC Anesthesiology. We described in detail all changes referring to your comment in the order mentioned in the review. Sincerely Dr Han

1 Authorship change form

We note the removal of an author- JYH since original submission of the manuscript. In line with COPE guidelines, BioMed Central requires written confirmation from all authors that they agree with any proposed changes in authorship of submitted manuscripts or published articles. In such cases, we use a standardised form which we would request you and your co-authors to complete. The change in authorship form can be found from the link on the following page: https://www.biomedcentral.com/submissions/editorial-policies#authorship

All instructions can be found on the form, please treat the 'current authorship' section as the original authorship. Please return the form within 14 days with all author signatures (including those newly added/removed). Please also ensure that the relevant authors change the statements in column 4 of the signature table to reflect whether their name is being added. When completed please send the form to the Editorial office (BMCseriesEditorial@biomedcentral.com) and cc lingling.tian@biomedcentral.com with subject 'Authorship change form + your manuscript ID’ : We apologize for this mistake. We found this error thanks to your comment. We confirm that all authors of the first draft are included in the final version.

2 Authors' contributions

Please update the Authors' Contributions in accordance with the change of the authors. Currently, the author JYH is still present in this section: In line with the answers of the No 1, all authors are included in the authors’ contribution.

3 In order to adhere to submission guideline, please rename the section heading 'Materials and methods' to 'Methods': According to the comment, Materials and methods was replaced with Methods.

4 Ethics approval
and consent to participate. It is stated 'Not applicable'. Ethics approval is applicable to your study as it involves animal work. In this section, please include the full name of the ethics committee (and the institute to which it belongs to) that approved the study and the committee’s reference number.

Thank you for the comment. “Institutional animal care and use committee (IACUC) of Seoul National University Bundang Hospital (No BA1204-103/030-01)” was added. 5 Availability of data and materials It is stated 'The datasets generated during and/or analysed during the current study are available in the figshare repository (https://doi.org/10.6084/m9.figshare.5287735).'- However, the weblink is not accessible. Please check the weblink. Alternatively, you may want to supply the Availability of Data and Materials statement as below: The datasets generated and analysed during the current study are available from the corresponding author on reasonable request.

The authors appreciate your advice. According to the comment, “The datasets generated and analyzed during the current study are available from the corresponding author on reasonable request” was added. 6 Acknowledgements Please include a section 'Acknowledgements' in Declarations. Please only use this section to acknowledge anyone who contributed towards the article who does not meet the criteria for authorship including anyone who provided professional writing services or materials. Authors should obtain permission to acknowledge from all those mentioned in the Acknowledgements section. If no acknowledgements are necessary then please state 'None'. None was added at the Acknowledgements section. 7 Upon resubmission, please make sure that your manuscript contains the following seven sections under the heading 'Declarations' at the end of the main text: (1) Ethics approval and consent to participate (2) Consent for publication (3) Availability of data and material (4) Competing interests (5) Funding: (6) Authors' contributions (7) Acknowledgements 8 Upon resubmission, Please remove any tracked changes or highlighting and include only a single clean copy of the manuscript. Please ensure the document is in the final form for publication; please upload only files that are to be published. If you wish to respond to these revision requests, please insert the information in to the designated input box only: We confirmed that any tracked changes or highlighting was removed throughout the manuscript.